

Citation: Wong CKH, Man KKC, Shi M, Chan EW, Ho CW, Tse ETY, et al. (2019) Intensification with dipeptidyl peptidase-4 inhibitor, insulin, or thiazolidinediones and risks of all-cause mortality, cardiovascular diseases, and severe hypoglycemia in patients on metformin-sulfonylurea dual therapy: A retrospective cohort study. PLoS Med 16(12): e1002999. https://doi.org/10.1371/journal. pmed.1002999

Academic Editor: Sanjay Basu, Harvard Medical School, UNITED STATES

Received: August 8, 2019

Accepted: November 21, 2019

Published: December 26, 2019

Copyright: © 2019 Wong et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data underlying the results presented in the study are available from Administrative Assessment of External Data Requests, Hong Kong Hospital Authority Head Office (contact via hacpaaedr@ha.org.hk).

Funding: This study was funded by the Health and Medical Research Fund Research Fellowship Scheme, Food and Health Bureau, Hong Kong SAR RESEARCH ARTICLE

Intensification with dipeptidyl peptidase-4 inhibitor, insulin, or thiazolidinediones and risks of all-cause mortality, cardiovascular diseases, and severe hypoglycemia in patients on metformin-sulfonylurea dual therapy: A retrospective cohort study

Carlos K. H. Wong^{1*}, Kenneth K. C. Man^{2,3}, Margaret Shi¹, Esther W. Chan², Chu Wa Ho¹, Emily T. Y. Tse¹, Ian C. K. Wong^{2,3}, Cindy L. K. Lam¹

1 Department of Family Medicine and Primary Care, Li Ka Shing Faculty of Medicine, The University of Hong Kong, Hong Kong SAR, China, 2 Centre for Safe Medication Practice and Research, Department of Pharmacology and Pharmacy, Li Ka Shing Faculty of Medicine, The University of Hong Kong, Hong Kong SAR, China, 3 Research Department of Policy and Practice, School of Pharmacy, University College London, London, United Kingdom

* carlosho@hku.hk

Abstract

Background

Although patients with type 2 diabetes mellitus (T2DM) may fail to achieve adequate hemoglobin A1c (HbA1c) control despite metformin-sulfonylurea (Met-SU) dual therapy, a thirdline glucose-lowering medication—including dipeptidyl peptidase-4 inhibitor (DPP4i), insulin, or thiazolidinedione (TZD)—can be added to achieve this. However, treatment effects of intensification with the medications on the risk of severe hypoglycemia (SH), cardiovascular disease (CVD), and all-cause mortality are uncertain. Study aim was to compare the risks of all-cause mortality, CVD, and SH among patients with T2DM on Met-SU dual therapy intensified with DPP4i, insulin, or TZD.

Methods and findings

We analyzed a retrospective cohort data of 17,293 patients with T2DM who were free from CVD and on Met-SU dual therapy and who were intensified with DPP4i (n = 8,248), insulin (n = 6,395), or TZD (n = 2,650) from 2006 to 2017. Propensity-score weighting was used to balance out baseline covariates across groups. Hazard ratios (HRs) for all-cause mortality, CVD, and SH were assessed using Cox proportional hazard models. Mean age of all patients was 58.56 ± 11.41 years. All baseline covariates achieved a balance across the 3 groups. Over a mean follow-up period of 34 months with 49,299 person-years, cumulative incidences of all-cause mortality, SH, and CVD were 0.061, 0.119, and 0.074, respectively. Patients intensified with insulin had higher risk of all-cause mortality (HR = 2.648, 95% confidence interval [CI] 2.367–2.963, p < 0.001; 2.352, 95% CI 2.123–2.605, p < 0.001) than

(Ref No. 02160087). The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: All authors have completed the ICMJE uniform disclosure form at www.icmje.org/coi_disclosure.pdf and declare: no support from any organization for the submitted work (except the research grants listed below and in funding sources section); no financial relationships with any organizations that might have an interest in the submitted work in the previous three years; no other relationships or activities that could appear to have influenced the submitted work. Dr CKHW reports receipt of research funding from the EuroQoL Group Research Foundation, the Hong Kong Research Grants Council, and the Hong Kong Health and Medical Research Fund. Dr KKCM reports receipt of CW Maplethorpe Fellowship and received personal fees from IQVIA Holdings. Inc., unlated to this work. Dr EWYC reports receipt of research funding from Bristol-Myers Squibb, Pfizer, Janssen, Takeda Pharmaceuticals, the Hong Kong Beat Drugs Fund Association, the Hong Kong Research Grants Council, and the Hong Kong Health and Medical Research Fund. Prof ICKW reports receipt of research funding from Wellcome Trust, United Kingdom; National Natural Science Fund of China, China; The Hong Kong Research Grants Council, The Research Fund Secretariat of the Food and Health Bureau, Narcotics Division of the Security Bureau of HKSAR, Hong Kong; Bristol-Myers Squibb, Pfizer, Bayer and Janssen, a Division of Johnson & Johnson Takeda, for work unrelated to this study. Prof CLKL reports receipt of research funding from the Hong Kong Health and Medical Research Fund, the Hong Kong Research Grants Council, and the Kerry Group and Kouk Foundation Endowed Primary Care Research Fund of the University of Hong Kong, unrelated to this study. No other disclosures were reported.

Abbreviations: ADA, American Diabetes Association; ASMD, absolute standardized mean difference; BMI, body mass index; CCI, Charlson comorbidity index; CI, confidence interval; CVD, cardiovascular disease; DBP, diastolic blood pressure; DPP4i, dipeptidyl peptidase-4 inhibitor; EASD, European Association for the Study of Diabetes; eGFR, estimated glomerular filtration rate; GLP1-RA, glucagon-like peptide-1 receptor agonist; HbA1c, hemoglobin A1c; HDL-C, highdensity lipoprotein cholesterol; HR, hazard ratio; ICD-9-CM, International Statistical Classification of Diseases and Related Health Problems, 9th those intensified with TZD and DPP4i, respectively. Insulin users had the greatest risk of SH (HR = 1.198, 95% CI 1.071–1.340, p = 0.002; 1.496, 95% CI 1.342–1.668, p < 0.001) compared with TZD and DPP4i users, respectively. Comparing between TZDs and DPP4i, TZDs were associated with a higher risk of SH (HR = 1.249, 95% CI 1.099–1.419, p < 0.001) but not all-cause mortality (HR = 0.888, 95% CI 0.776–1.016, p = 0.084) or CVD (HR = 1.005, 95% CI 0.915–1.104, p = 0.925). Limitations of this study included the lack of data regarding lifestyle, drug adherence, time-varying factors, patients' motivation, and cost considerations. A limited duration of patients intensifying with TZD might also weaken the strength of study results.

Conclusions

Our results indicated that, for patients with T2DM who are on Met-SU dual therapy, the addition of DPP4i was a preferred third-line medication among 3 options, with the lowest risks of mortality and SH and posing no increased risk for CVD events when compared to insulin and TZD. Intensification with insulin had the greatest risk of mortality and SH events.

Author summary

Why was this study done?

- For those who fail dual therapy of metformin and sulfonylurea (Met-SU) after treatment initiation for 3 months, treatment intensification with a third-line glucose-lowering medication is considered as logical stepwise approach of pharmacotherapy.
- Current guidelines endorsed the addition of thiazolidinedione (TZD), dipeptidyl peptidase-4 inhibitors (DPP4i's), or basal insulin regimen as third-line options.
- This study directly compared the risks of cardiovascular disease (CVD) and all-cause mortality among type 2 diabetes mellitus (T2DM) patients receiving DPP4i, TZD, or insulin as a third-line medication.

What did the researchers do and find?

- We analyzed a population-wide retrospective cohort with data collected from the Hong Kong Hospital Authority database from January 2006 to December 2017.
- Patients who were free from CVD and on Met-SU dual therapy, and whose drug treatment was intensified with DPP4i (n = 8,248), insulin (n = 6,395), or TZD (n = 2,650), were included.
- Patients intensified with insulin had higher risk of all-cause mortality than those intensified with TZD and DPP4i, respectively.
- Insulin users had the greatest risk of severe hypoglycemia (SH) compared with TZD and DPP4i users, respectively, whereas TZD users were associated with a higher risk of SH than DPP4i users.

Revision, Clinical Modification; ICPC-2, International Classification of Primary Care, Version 2; IR, incidence rate; LDL-C, low-density lipoprotein cholesterol; Met-SU, metforminsulfonylurea; MICE, multiple imputation by chained equations; NICE, National Institute for Health and Care Excellence; NPH, Neutral Protamine Hagedorn; RCT, randomized controlled trial; SBP, systolic blood pressure; SGLT2, sodium-glucose co-transporter-2; SH, severe hypoglycemia; SHR, subdistribution hazard ratio; T2DM, type 2 diabetes mellitus; TZD, thiazolidinedione.

What do these findings mean?

In this population-based retrospective cohort study of patients with T2DM after failure of Met-SU dual therapy, TZD was associated with greater risk of SH than DPP4i, whereas insulin was associated with the greatest risks of all-cause mortality and SH. In this cohort, DPP4i had a lower risk of all-cause mortality compared with insulin, and a lower risk of SH compared with TZD and insulin. Caution needs to be exercised when using TZD or insulin because of its high risk for death or SH.

Introduction

Type 2 diabetes mellitus (T2DM) is a chronic condition with a potential risk of developing long-term complications without adequate glycemic control [1]. The combined use of glucoselowering medications in addition to lifestyle modification and monotherapy are often required to achieve personalized glycemic targets in patients with T2DM. For those who had inadequate glycemic control under dual therapy of metformin and sulfonylurea (Met-SU) after treatment initiation for 3 months, treatment intensification with a third-line glucose-lowering medication is considered as logical stepwise approach of pharmacotherapy [2]. The updated Position Statement [3] by the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD) endorsed the addition of one of the following glucose-lowering medications as a third-line option when optimal glycemic control is not achieved after 3 months of dual therapy: thiazolidinedione (TZD), dipeptidyl peptidase-4 inhibitors (DPP4i), sodium-glucose co-transporter-2 (SGLT2) inhibitors, glucagon-like peptide-1 receptor agonists (GLP1-RAs), or basal insulin regimen. This clinical practice has also been recommended in the National Institute for Health and Care Excellence (NICE) guideline in 2015 whereby triple therapy should be considered when dual therapy has not continued to control hemoglobin A1c (HbA1c) to below an individual's target [4]. The consensus report published in 2018 by the ADA and the EASD [5], which was reinstated in 2019 by the ADA [6], advocated the use of TZD following Met-SU therapy when cost is a major issue in patients without existing cardiovascular disease (CVD). Furthermore, the cost-effectiveness of third-line medications was assessed by the Canadian Agency for Drugs and Technologies in Health in 2013 [7]. The addition of basal insulin to Met-SU therapy was the most favorable treatment regimen in terms of cost-effectiveness estimates [7]. Nevertheless, patient preferences, risks and benefits of medications, age, initial HbA1c level, and other patient-specific factors are determinants of decisions about third-line drug options [8].

Previous literature [9,10] has shown that DPP4i's were not associated with cardiovascular adverse effects and can be used safely even in the elderly. Nevertheless, caution needs be exercised when prescribing saxagliptin (DPP4i) and alogliptin (DPP4i) to patients who are at high risk of heart failure [3,9,11,12]. The TZD class is highly effective in reducing HbA1c level and is associated with a low risk of hypoglycemia [12]. Despite the risk of fluid retention with the likelihood of heart failure, pioglitazone (TZD) has been shown to confer cardiovascular benefits, and it can be given in a moderate dose (\leq 30 mg) for side effects to be mitigated [12]. On the other hand, rosiglitazone (TZD) was subsequently suspended from the European market because of its associated high risk of myocardial infarction and heart failure [3,11]. With regard to insulin, a vicious cycle is likely to be present between persistently high HbA1c level and insulin resistance, whereby an increased dose is necessary to control hyperglycemia leading to further weight gain, resulting in a further increase in CVD risk [11]. Compared to

DPP4i, insulin was reported to have a higher risk of CVD and all-cause mortality [10]. Hypoglycemia is a known risk factor for cardiovascular adverse events, and the use of insulin may be linked to an increase in life-threatening hypoglycemia events requiring external assistance for recovery [13].

Several network meta-analyses [13–15] of randomized controlled trials (RCTs) with a duration of pharmacotherapy of less than 1 year synthesized the short-term intermediate outcomes of each third-line drug option in patients with dual therapy failure. Review of the RCTs with treatment duration between 6 and 12 months demonstrated that the addition of a third-line therapy to dual therapy resulted in improvement in HbA1c from -0.56% to -0.94% and was associated with greater risks of hypoglycemia [16]. Furthermore, treatment intensification with DPP4i to Met-SU dual therapy resulted in significantly less weight gain compared with adding insulin or TZD as the third-line medication [14,17]. However, previous analyses lacked large-scale long-term follow-up regarding the impact of third-line drug options on CVD and mortality outcomes. Notably, when DPP4i was added to Met-SU dual therapy, additional cardiovascular benefits with a lower risk of all-cause mortality were demonstrated in a nationwide retrospective cohort study, in comparison to other oral glucose-lowering medications (i.e., acarbose, meglitidine) [18]. In addition, those on DPP4i as a third-line treatment had a significantly lower risk of heart failure compared with those on a meglitidine add-on regimen. However, no statistically significant difference was seen in stroke risk reduction compared with TZD [18].

The present study examined the treatment effect of intensification with DPP4i, insulin, or TZD on the risk of severe hypoglycemia (SH), CVD, and all-cause mortality among patients with T2DM on Met-SU dual therapy. We hypothesized that patients receiving DPP4i as triple therapy would have lower risks of all-cause mortality, SH, and CVD compared with those receiving insulin or TZD. Moreover, it was hypothesized that intensification with insulin as a third-line medication would be most detrimental to outcome events among the 3 medications.

Methods

Ethics approval of this study was granted by the Institutional Review Board of the University of Hong Kong/Hospital Authority Hong Kong West Cluster (Reference No. UW 16–1018).

Data source description

We analyzed the population-based retrospective cohort from the Hong Kong Hospital Authority administrative database in the Hong Kong adult diabetes population from January 1, 2006, to December 31, 2017. The Hospital Authority database has been extensively used for conducting high-quality large population-based studies [19–22]. Documented diabetes mellitus diagnosis was defined as the International Classification of Primary Care, Version 2 (ICPC-2) codes T89/T90 or International Statistical Classification of Diseases and Related Health Problems, 9th Revision, Clinical Modification (ICD-9-CM) codes 250.x. The database contains comprehensive individual patient-level information on prescription and dispensing of glucose-lowering medication, serial readings of anthropometric and laboratory variables, and presence of comorbidities as classified based on the ICD-9-CM or ICPC-2 diagnosis codes. A prospective study protocol or analysis plan was not available when designing this study. This study's statistical analysis plan was developed in March 2019. Changes in the analysis, including those sensitivity and additional analyses made in response to reviewers' comments, took place in October 2019.

Identification of study population

All glucose-lowering medication dispensed in pharmacy departments managed under the Hospital Authority during the study period were identified for this study. Information on the date of drug prescribing and dispensing, dosage unit, and quantity was recorded. We included patients who were on dual therapy of Met-SU in the study period and were subsequently intensified with one of the following glucose-lowering medications: DPP4i, insulin, or TZD. Patients were excluded who met the following criteria: less than 18 years old, had type 1 diabetes mellitus, had no diabetes mellitus diagnosis code, had a CVD event occur before the initiation of the third-line medication, received first glucose-lowering medications within 180 days after third-line initiation, and received other glucose-lowering medication drug classes before Met-SU was commenced. Baseline date of eligible patients was defined as the date of initiating third-line medication. Patients were observed from the baseline date until the occurrence of study outcome, death from any cause, or being censored at the last follow-up date, whichever came first.

Outcome measures

Our study outcomes were all-cause mortality, SH, and composite CVD (acute myocardial infarction, other ischemic heart disease, congestive heart failure, stroke, and peripheral vascular disease). Both the SH and CVD events were identified by the diagnosis codes of the ICD-9-CM and the ICPC-2. All the ICD-9-CM and ICPC-2 diagnosis codes for comorbidities and event outcomes are listed in <u>S1 Table</u>.

Baseline covariates

The baseline covariates included age, gender, and clinical characteristics such as body weight, body mass index (BMI), HbA1c, systolic blood pressure (SBP), diastolic blood pressure (DBP), total cholesterol levels, low-density lipoprotein cholesterol (LDL-C), high-density lipoprotein cholesterol (HDL-C), Charlson comorbidity index (CCI), history of SH, and duration of diabetes mellitus drug dispensed before intensification with third-line medication (i.e., the time between the first glucose-lowering medication prescription and the first third-line medication prescription).

Statistical analysis

Patients were grouped according to their third-line medication. Baseline characteristics were described respectively (mean \pm SD for continuous variables, N [%] for categorical variables).

To address missing baseline data, multiple imputation by chained equations (MICE) [23] was used for 3 groups. BMI, HbA1c, SBP, DBP, LDL-C, total cholesterol, HDL-C, serum creatinine, triglyceride, and fasting glucose were imputed by other parameters such as gender, age, duration between first-line medication and third-line medication, history of SH, and CCI. Model parameters were estimated from multiple imputed data and then used to obtain multiple-imputation linear predictions by applying Rubin's combination rules observation wise to the completed-data predictions [23]. Propensity-score weighting was applied using the predictions obtained after MICE.

To minimize the outcome bias due to discrepancy in baseline covariates, inverse probability of treatment weights (IPTW) using the propensity score was applied to balance covariates across the 3 groups. A multinomial logistics regression model was performed to calculate the propensity scores of each patient in group and included the covariates of age, gender, CCI, and

duration between first-line medication and third-line medication. Duration of patient on Met-SU dual therapy was calculated to account for immortal time bias [24] and immortal persontime exposed to Met-SU in each group. The IPTW using the propensity scores was implemented using a user-written command marginal mean weighting through stratification. The lowest and highest 1% (corresponding to the 1st and 99th percentiles) propensity-score weights in each group were removed to trim extreme weights [25]. By repeating these steps 5 times, propensityscore weightings and treatment effects estimated from each imputed dataset were combined to obtain overall estimates of treatment effect. In the context of IPTW, the multiple imputation followed by pooling treatment effects estimates across imputed datasets is the preferred approach [26]. After the propensity-score weighting, the balance of baseline covariates between the groups was further assessed using the absolute standardized mean difference (ASMD). All maximum pairwise ASMDs less than 0.2 implied optimal balance between the groups [27].

Incidence rates (IRs) of each outcome event for each group were estimated using the total number of patients with event occurrence during the follow-up period divided by personyears at risk. A Cox proportional hazards regression model was used to examine the association between the third-line medications and incidence of events. Hazard ratios (HRs) and 95% confidence intervals (CIs) were reported for each treatment group in the regression model. A log-rank test was used to compare the equality of the survival curves between the groups. Predictive accuracy of Cox models was assessed and compared using Harrell's discrimination Cindex, ranging from 0 to 1. A value of 0.5 indicates no predictive discrimination, and values of 0 or 1.0 indicate perfect separation of patients. Proportional hazards assumptions were confirmed through the Schoenfeld residuals test. Goodness of fit of the Cox regression model was assessed using the Akaike information criterion and Bayesian information criterion.

A sensitivity analysis was conducted to include pioglitazone only in the TZD drug class and exclude rosiglitazone because it had already been taken off the market in many countries. Likewise, the effects of basal insulin (Neutral Protamine Hagedorn [NPH] insulin and long-acting insulin) within the insulin class in lowering the risks of all-cause mortality, SH, and CVD were assessed in a sensitivity analysis. The competing risk for mortality was accounted for by the analysis of SH and CVD events, and complete-case analysis in sensitivity analyses. We calculated the E-values as a sensitivity analysis to quantify the potential for unmeasured confounding bias on observed treatment–outcome association [28,29]. The E-value is the minimum strength of association required between an unmeasured confounder and treatment and between confounder and outcome—conditional on measured covariates—to negate the observed treatment-outcome association [28,29]. E-values for each outcome were calculated using an online calculator [30].

All statistical analyses were performed using STATA version 13.0 (StataCorp LP, College Station, Texas). All significance tests were two-tailed, and p < 0.05 was taken to indicate statistical significance. Statistical analyses were conducted by two coauthors (CKHW and CWH) and cross-checked for quality assurance.

Results

The selection process of the cohort group is outlined in the flowchart in Fig 1. In total, 17,293 eligible patients were included in the current analysis. Among all patients, a majority (47.7%) received DPP4i as their third-line medication, followed by insulin (37.0%) and TZD (15.3%).

Patient characteristics

Table 1 illustrates the baseline characteristics of patients according to their treatment groups before weighting. All the maximum pairwise ASMDs of the characteristics were less than 0.2



Fig 1. Enrolment of patients who had failure of Met-SU dual therapy and received DPP4i, insulin, or TZD as third-line medications. DM, diabetes mellitus; DPP4i, dipeptidyl peptidase-4 inhibitor; Met-SU, metformin-sulfonylurea; TZD, thiazolidinedione.

https://doi.org/10.1371/journal.pmed.1002999.g001

after weighting, implying that all covariates achieved a balance in baseline covariates across the 3 groups. In this cohort, the mean age of all participants was 58.56 years, and mean BMI was 28.45 kg/m². Mean values of SBP and DBP were 132.48 and 76.89 mmHg, respectively. Details of baseline characteristics in each group after weighting are listed in S2 Table.

IRs

Table 2 depicts the cumulative incidence and IRs of all-cause mortality, SH, and CVD across the follow-up period for patients treated with insulin, DPP4i, and TZD as part of the triple therapy. Over a mean follow-up period of 34 months with 49,299 person-years, cumulative incidences of all-cause mortality, CVD, and SH were 0.061, 0.074, and 0.119, respectively. The mean follow-up period of our cohort ranged from 33 to 34 months across outcome events. Upon weighting, patients intensified with insulin had the most incidences of all-cause mortality (IR = 2.748/100 person-years), whereas the patients intensified with TZD had the most incidences of SH (IR = 4.896/100 person-years) and CVD (IR = 2.583/100 person-years).

Risk of SH, CVD, and all-cause mortality

Fig 2 depicts the Kaplan Meier survival curves for all-cause mortality, SH, and CVD events by treatment group.

Table 3 compares the HR of DPP4i, insulin, and TZD against each other. Risk of SH for insulin was increased by about 1.5-fold (HR = 1.496, 95% CI 1.342-1.668, p < 0.001) relative to DPP4i, followed by approximately 1.2-fold (HR = 1.249, 95% CI 1.099–1.419, p < 0.001) for TZD compared to DPP4i. Compared with the risk of SH, the risk of all-cause mortality for insulin was increased further to 2.648 (95% CI 2.367–2.963, p < 0.001) and 2.352 (95% CI 2.123-2.605, p < 0.001) relative to TZD and DPP4i, respectively. On the other hand, the difference in risk of all-cause mortality between TZD and DPP4i was insignificant (HR = 0.888, 95%) CI 0.776–1.016, p = 0.084). Similarly, the differences in risk of CVD between insulin and DPP4i (HR = 0.970, 95% CI 0.893–1.053, p = 0.476), between TZD and DPP4i (HR = 1.005, 95% CI 0.915–1.104, p = 0.928), and between insulin and TZD (HR = 0.965, 95% CI 0.883– 1.056, p = 0.446) were insignificant. To compare among the 3 medications and other glucoselowering medications, an additional analysis was conducted (S3 Table). Other glucose-lowering medications showed a significantly higher risk for mortality, SH, and CVD than DPP4i (HR = 1.288, 95% CI 1.154 - 1.439, p < 0.001; 2.414, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.180 - 2.674, p < 0.001; 1.834, 95% CI 2.674, p < 0.001; 1.834, 1.834, 1.834, 1.834, 1.834, 1.834, 1.834, 1.834, 1.834, 1.834, 1.834, 1.834, 1.834, 1.CI 1.702–1.977, p < 0.001, respectively) and TZD (HR = 1.462, 95% CI 1.301–1.643, p < 0.001; 1.932, 95% CI 1.739–2.146, p < 0.001; 1.889, 95% CI 1.741–2.050, p < 0.001, respectively). When comparing other medications with insulin, the risk of SH and CVD increased while that of mortality decreased. Overall, insulin was associated with the greatest risks in all cause-mortality and SH, followed by TZD and DPP4i, whereas there were no significant differences in risk of CVD across the 3 groups. DPP4i had the lowest risk of all-cause mortality and SH events as a third-line agent for intensification of diabetes management.

Sensitivity analysis

Given that rosiglitazone and pioglitazone belong to the drug class of TZD, when those who received rosiglitazone medications were excluded in the sensitivity analysis, it became evident

Table 1. Baseline characteristics of patients intensifying with DPP4i, insulin or TZD, as third-line medication.

					Maximum pairwise ASMD	
	Total	DPP4i	Insulin	TZD	Before weighting	After weighting
General information						
Total number of participants	17,293	8,248	6,395	2,650		
Age (years), mean ± SD	58.56 ± 11.41	59.22 ± 10.98	58.05 ± 12.27	57.73 ± 10.38	0.138	0.072
Gender, <i>n</i> (%)					0.021	0.021
Female	7,959 (46.02%)	3,812 (46.22%)	2,950 (46.13%)	1,197 (45.17%)		
Male	9,334 (53.98%)	4,436 (53.78%)	3,445 (53.87%)	1,453 (54.83%)		
Clinical parameter						
Laboratory result, mean ± SD						
HbA1c, %	8.71 ± 1.59	8.49 ± 1.25	9.16 ± 1.97	8.31 ± 1.22	0.476*	0.069
SBP, mmHg	132.48 ± 16.07	132.85 ± 15.59	133.10 ± 17.22	129.84 ± 14.34	0.199	0.021
DBP, mmHg	76.89 ± 9.67	76.96 ± 9.78	76.97 ± 9.87	76.43 ± 8.76	0.057	0.016
LDL-C, mmol/L	2.42 ± 0.77	2.42 ± 0.74	2.49 ± 0.83	2.25 ± 0.68	0.314*	0.097
HDL-C, mmol/L	1.13 ± 0.29	1.13 ± 0.27	1.14 ± 0.31	1.12 ± 0.27	0.075	0.028
BMI, kg/m ²	28.45 ± 4.06	28.60 ± 4.04	28.15 ± 4.10	28.73 ± 4.01	0.142	0.032
Waist, cm	96.10 ± 19.15	95.99 ± 9.85	96.07 ± 24.58	96.50 ± 25.54	0.033	0.076
TC, mmol/L	4.38 ± 0.94	4.37 ± 0.89	4.48 ± 1.02	4.15 ± 0.84	0.340*	0.101
Triglyceride, mmol/L	1.87 ± 1.30	1.86 ± 1.24	1.93 ± 1.42	1.78 ± 1.20	0.113	0.054
Creatinine (serum), µmol/L	89.04 ± 67.67	85.02 ± 51.97	98.36 ± 92.28	79.05 ± 23.79	0.246*	0.054
eGFR, mL/min/1.73 m ²	86.09 ± 28.82	86.84 ± 27.52	84.48 ± 32.80	87.65 ± 21.40	0.106	0.095
Fasting glucose, mmol/L	9.60 ± 3.02	9.36 ± 2.63	10.18 ± 3.53	8.96 ± 2.57	0.373*	0.084
Prior severe hypoglycemia, n (%)	1,299 (7.51%)	456 (5.53%)	741 (11.59%)	102 (3.85%)	0.293*	0.011
Duration between first-line medication and third-line medication (years), mean ± SD	5.54 ± 2.81	5.81 ± 2.71	4.90 ± 2.73	6.26 ± 3.00	0.483*	0.068
Duration of DM (years), mean ± SD	5.43 ± 2.97	5.69 ± 2.87	4.75 ± 2.89	6.36 ± 3.10	0.546*	0.063
Duration of DM (years), <i>n</i> (%)					0.309*	0.080
\leq 5 years	7,210 (46.57%)	2,994 (42.40%)	3,330 (55.66%)	886 (36.34%)		
5–10 years	7,200 (46.50%)	3,579 (50.68%)	2,427 (40.56%)	1,194 (48.97%)		
>10 years	1,073 (6.93%)	489 (6.92%)	226 (3.78%)	358 (14.68%)		
CCI, n (%)					0.479*	0.045
1 or 2	2,662 (15.39%)	1,149 (13.93%)	1,112 (17.39%)	401 (15.13%)		
3	4,535 (26.22%)	2,165 (26.25%)	1,520 (23.77%)	850 (32.08%)		
4	4,443 (25.69%)	2,243 (27.19%)	1,386 (21.67%)	814 (30.72%)		
5	2,793 (16.15%)	1,527 (18.51%)	885 (13.84%)	381 (14.38%)		
6 or above	2,860 (16.54%)	1,164 (14.11%)	1,492 (23.33%)	204 (7.70%)		

*Imbalance covariate if maximum pairwise ASMD \geq 0.2.

Abbreviations: ASMD, absolute standardized mean difference; BMI, body mass index; DBP, diastolic blood pressure; DPP4i, dipeptidyl peptidase-4 inhibitor; CCI, Charlson comorbidity index; DM, diabetes mellitus; eGFR, estimated glomerular filtration rate; HbA1c, hemoglobin A1c; HDL-C, high-density lipoprotein cholesterol; LDL-C, low-density lipoprotein cholesterol; SBP, systolic blood pressure; SD, standard derivation; SH, severe hypoglycemia; TC, total cholesterol; TZD, thiazolidinedione

https://doi.org/10.1371/journal.pmed.1002999.t001

that the risk of SH for pioglitazone users becomes insignificant compared with DPP4i (HR = 1.102, 95% CI = 0.956–1.271, p = 0.180). In contrast to insulin class, basal insulin has a lower risk of mortality and SH in the comparison group with DPP4i, and a statistically significant lower risk of SH (HR = 0.830, 95% CI 0.750–0.918, p < 0.001) and CVD (HR = 0.839, 95% CI 0.773–0.912, p < 0.001) than TZD. In the meantime, prandial insulin showed a relatively higher risk for mortality, SH, and CVD compared with DPP4i, and the hazard of CVD

	Before weighting							After weighting	
Event	Cumulative incidence		Crude incidence rate (cases/100 person- years)			Median follow-up periods (months)	Mean follow-up periods (months)	Incidence rate (cases/100 person-years)	
	Cases with event	Rate	Estimate	95% CI*	Person-years			Estimate	95% CI*
Total (N = 17,293)									
All-cause mortality	1,057	0.061	2.144	(2.017-2.277)	49,299.25	28	34	1.805	(1.733-1.878)
SH	2,065	0.119	4.698	(4.497-4.905)	43,959.33	26	33	4.120	(4.006-4.235)
CVD	1,276	0.074	2.698	(2.552-2.850)	47,293.83	26	33	2.445	(2.360-2.532)
DPP4i as third-line (N = 8,248)									
All-cause mortality	228	0.028	1.035	(0.905-1.178)	22,032.92	29	32	1.120	(1.024–1.222)
SH	688	0.083	3.333	(3.088-3.591)	20,642.92	28	32	3.738	(3.554-3.928)
CVD	477	0.058	2.225	(2.030-2.434)	21,436.42	28	31	2.394	(2.251-2.544)
Insulin as third-line (N = 6,395)									
All-cause mortality	796	0.124	3.439	(3.204-3.686)	23,147.42	40	43	2.748	(2.609–2.891)
SH	1,236	0.193	6.379	(6.028-6.744)	19,377.25	37	41	3.957	(3.782-4.136)
CVD	732	0.114	3.361	(3.122-3.613)	21,781.92	36	41	2.402	(2.270-2.538)
TZD as third-line $(N = 2,650)$									
All-cause mortality	33	0.012	0.801	(0.551-1.125)	4,118.92	13	19	1.200	(1.086-1.323)
SH	141	0.053	3.579	(3.013-4.221)	3,939.17	14	19	4.896	(4.650-5.148)
CVD	67	0.025	1.644	(1.274–2.088)	4,075.50	14	18	2.583	(2.411-2.764)

Table 2. Number and incidence rate of all-cause mortality, SH, and CVD events.

*The 95% CI was constructed based on Poisson distribution.

Abbreviations: CI, confidence interval; CVD, cardiovascular disease; DPP4i, dipeptidyl peptidase-4 inhibitor; SH, severe hypoglycemia; TZD, thiazolidinedione

https://doi.org/10.1371/journal.pmed.1002999.t002

was significantly decreased when compared with TZD (HR = 1.017, 95% CI 0.932–1.110, p = 0.724) (Table 3). Accounting for the competing risk of death in the CVD outcome analysis, results were in line with those in the main analysis. Insulin was associated with a lower risk of



Fig 2. Kaplan Meier survival curves for all-cause mortality, SH, and CVD for T2DM patients with DPP4i, insulin, or TZD as third-line medications after propensity-score weighting. CVD, cardiovascular disease; DPP4i, dipeptidyl peptidase-4 inhibitor; SH, severe hypoglycemia; T2DM, type 2 diabetes mellitus; TZD, thiazolidinedione.

https://doi.org/10.1371/journal.pmed.1002999.g002

	TZD (versus DPP4i)			Insulin (versus TZD)			Insulin (versus DPP4i)		
HR	95% CI	<i>p</i> -value	HR	95% CI	<i>p</i> -value	HR	95% CI	<i>p</i> -value	
0.888	(0.776-1.016)	0.084	2.648	(2.367-2.963)	< 0.001*	2.352	(2.123-2.605)	< 0.001*	
1.249	(1.099–1.419)	$< 0.001^{*}$	1.198	(1.071-1.340)	0.002*	1.496	(1.342–1.668)	< 0.001*	
1.005	(0.915-1.104)	0.928	0.965	(0.883-1.056)	0.446	0.970	(0.893-1.053)	0.476	
litazone onl	y in TZD drug class								
Event TZD (versus DPP4i)		i)	Insulin (versus TZD)			Insulin (versus DPP4i)			
HR	95% CI	<i>p</i> -value	HR	95% CI	<i>p</i> -value	HR	95% CI	<i>p</i> -value	
0.917	(0.794–1.058)	0.237	2.586	(2.283-2.928)	< 0.001*	2.371	(2.140-2.626)	< 0.001*	
1.102	(0.956-1.271)	0.180	1.383	(1.215–1.574)	< 0.001*	1.524	(1.367–1.700)	< 0.001*	
0.925	(0.834–1.026)	0.141	1.080	(0.976-1.195)	0.136	0.999	(0.919–1.085)	0.981	
l insulin on	ly in insulin class								
	TZD (versus DPP4i)			Insulin (versus TZD)			Insulin (versus DPP4i)		
HR	95% CI	p-value	HR	95% CI	p-value	HR	95% CI	p-value	
0.902	(0.791-1.027)	0.120	1.631	(1.461–1.822)	< 0.001*	1.471	(1.323–1.635)	< 0.001*	
1.637	(1.456-1.841)	$< 0.001^{*}$	0.830	(0.750-0.918)	< 0.001*	1.358	(1.220-1.513)	< 0.001*	
1.205	(1.103–1.317)	$< 0.001^{*}$	0.839	(0.773-0.912)	< 0.001*	1.011	(0.933-1.097)	0.796	
dial insulin	only in insulin class								
	TZD (versus DPP4i)			Insulin (versus TZD)			Insulin (versus DPP4i)		
HR	95% CI	<i>p</i> -value	HR	95% CI	<i>p</i> -value	HR	95% CI	<i>p</i> -value	
1.309	(1.147-1.493)	< 0.001*	3.250	(2.936-3.598)	< 0.001*	4.254	(3.829-4.726)	< 0.001*	
1.620	(1.432–1.832)	$< 0.001^{*}$	1.784	(1.617–1.969)	< 0.001*	2.890	(2.604-3.207)	< 0.001*	
1.312	(1.192–1.443)	$< 0.001^{*}$	1.017	(0.932-1.110)	0.724	1.334	(1.222–1.455)	< 0.001*	
unting for c	ompeting risk of dea	th							
	TZD (versus DPP4i)		Insulin (versus TZD)			Insulin (versus DPP4i)			
SHR	95% CI	p-value	SHR	95% CI	p-value	SHR	95% CI	p-value	
1.264	(1.115–1.432)	< 0.001*	1.153	(1.031-1.291)	0.013*	1.458	(1.306-1.626)	< 0.001*	
1.015	(0.926-1.113)	0.758	0.927	(0.849-1.012)	0.092	0.941	(0.866-1.023)	0.155	
	HR 0.888 1.249 1.005 litazone onl HR 0.917 1.102 0.925 l insulin on HR 0.902 1.637 1.205 dial insulin HR 1.309 1.620 1.312 unting for c SHR 1.264 1.015	TZD (versus DPP4 HR 95% CI 0.888 (0.776–1.016) 1.249 (1.099–1.419) 1.005 (0.915–1.104) litazone only in TZD drug class TZD (versus DPP4 HR 95% CI 0.917 (0.794–1.058) 1.102 (0.956–1.271) 0.925 (0.834–1.026) <i>insulin only in insulin class</i> TZD (versus DPP4 HR 95% CI 0.902 (0.791–1.027) 1.637 (1.456–1.841) 1.205 (1.103–1.317) dial insulin only in insulin class TZD (versus DPP4 HR 95% CI 1.309 (1.147–1.493) 1.620 (1.432–1.832) 1.312 (1.192–1.443) unting for competing risk of dea TZD (versus DPP4 SHR 95% CI 1.264 (1.115–1.432) 1.015 (0.926–1.113)	TZD (versus DPP4i) HR 95% CI p -value 0.888 (0.776-1.016) 0.084 1.249 (1.099-1.419) <0.001*	TZD (versus DPP4i)HR95% CI p -valueHR0.888(0.776–1.016)0.0842.6481.249(1.099–1.419)<0.001*	TZD (versus DPP4i)Insulin (versus TZ)HR95% CI p -valueHR95% CI0.888(0.776-1.016)0.0842.648(2.367-2.963)1.249(1.099-1.419)<0.001*	TZD (versus DPP4i)Insulin (versus TZD)HR95% CI p -valueHR95% CI p -value0.888(0.776-1.016)0.0842.648(2.367-2.963) $<0.001^*$ 1.249(1.099-1.419) $<0.001^*$ 1.198(1.071-1.340)0.002^*1.005(0.915-1.104)0.9280.965(0.883-1.056)0.446Iteration of the transmitter of the transmitter of the transmitter of trans	TZD (versus DPP4i)Insulin (versus TZD)Insulin (versus TZD)HR95% CI p -valueHR0.888(0.776-1.016)0.0842.648(2.367-2.963)<0.001*	TZD (versus DPP4i)Insulin (versus TZD)Insulin (versus DPIHR95% CI p -valueHR95% CI p -valueHR95% CI0.88(0.76-1.016)0.0842.648(2.367-2.963)<0.01*	

Table 3. Hazard ratio of all-cause mortality, SH, and CVD events.

*Statistically significant at p < 0.05.

Abbreviations: CI, confidence interval; CVD, cardiovascular disease; DPP4i, dipeptidyl peptidase-4 inhibitor; HR, hazard ratio; SH, severe hypoglycemia; SHR, subdistribution hazard ratio; TZD, thiazolidinedione

https://doi.org/10.1371/journal.pmed.1002999.t003

SH compared with TZD (subdistribution hazard ratio [SHR] = 1.153, 95% CI 1.031–1.291, p = 0.013) and DPP4i (SHR = 1.458, 95% CI 1.306–1.626, p < 0.001) when accounting for the competing risk of death. The E-values as a sensitivity analysis for assessing unmeasured confounding bias were calculated for the HR for all-cause mortality, CVD, and SH outcomes (S4 Table). It was unlikely that an unmeasured or unknown confounder would have greater effect on the outcomes than these known risk factors by having a HR exceeding those E-values (S5 Table).

Discussion

In more recent years, promising new classes of oral dose forms have become options to add on to the dual combination of Met-SU when required [31]. The present study compared the risks of all-cause mortality, SH, and CVD among patients with T2DM who were on Met-SU dual therapy and intensified with DPP4i, insulin, or TZD. The addition of insulin as a third-line medication was found to have the highest incidence and risk of all-cause mortality, as well as an increased risk of SH. These findings are consistent with our hypothesis that patients with

T2DM initiated with insulin as a third-line medication could be at higher risk of SH and allcause mortality. A possible explanation of an increased mortality rate is the potential that insulin may increase the risk of developing cancer and atherosclerotic vascular diseases due to its atherogenic and mitogenic effects [32]. However, interesting results showed a lower risk of CVD compared with that of TZD. In the meantime, insulin remains an effective, potent glucose-lowering agent with an overall established safety record and is thus considered as part of a combination therapy when hyperglycemia is severe and poorly controlled with the use of oral agents alone [12]. When the use of basal insulin was assessed for the risks of all-cause mortality, SH, and CVD events in sensitivity analysis, basal insulin was shown to be associated with further reduced risks of all outcomes compared to insulin class. Such findings echoed Canadian guidelines that initiation of a single daily dose of insulin NPH under the basal insulin regimen has been recommended since 2013 [7].

DPP4i's, though ranked the best in number of incidences of all outcomes, were found to be associated with the lowest risk of mortality and SH outcomes. The latest clinical practice guideline "Optimal Use Recommendations for second- and third-line therapy for patients with type 2 diabetes" has recommended that DPP4i be used when patients are unable to use insulin as the third-line medication [7]. It was surprising to see that there was no difference in the risk of CVD between DPP4i and insulin. DPP4i was not associated with a lower risk of CVD than TZD, echoing the notion that DPP4i is known to be non-superior in its cardiovascular protective effects [9]. Notably, among the RCTs of TZD, there was a lack of comparison with other glucose-lowering medications other than with placebos [9]. This confirms the novelty of our study in addressing the research gap in terms of the limited evidence in the area of studying the risks of all-cause mortality, SH, and CVD across third-line medications, specifically DPP4i, insulin, and TZD. Our study further highlighted the importance of a population-based study whereby having a large sample size allows valid conclusions to be drawn for the assessment of risk of all-cause mortality, SH, and CVD for third-line medications for patients with T2DM.

While those using TZD were at greater risk of SH than those using DPP4i (HR = 1.249, 95% CI 1.099–1.419, p < 0.001), the risks of all-cause mortality and CVD were not statistically significant. As a subgroup of TZDs, pioglitazone was found to be associated with further decrease in the risk of SH in sensitivity analysis. There is a need for careful monitoring for baseline risks in patients intensifying with TZD and insulin, as well as careful assessment regarding the choice of a new add-on third-line agent for T2DM management.

Collectively, findings of this study provide new insights in understanding the risks of important classes of glucose-lowering agents as third-line medications and prove low risks of all-cause mortality and SH events associated with DPP4i, although insulin and TZD are highly effective glucose-lowering agents. The findings from this study should be considered when making clinical decisions about third-line glucose-lowering medications.

Limitations

Our study had some limitations. Lifestyle risk factors and issues with drug adherence were not captured in the database and could not be assessed. Therefore, it was not possible to include these factors in the propensity-score weighting. However, the likelihood that unmeasured confounders could affect the treatment–outcome relationship seemed unlikely, as indicated by E-values in sensitivity analysis. In addition, time-varying factors—such as changes in HbA1c, blood pressure, and lipid profile—were also not taken into account in the propensity-score weighting and subsequent multivariable analyses. These could potentially exert an influence on the risk of cardiovascular adverse effects and reduce the validity of results. Moreover, specific individual factors such as a patient's motivation and cost considerations were not

considered; however, propensity-score weighting was applied to other biomedical factors, including age and comorbidities. Furthermore, a limited duration of patients intensifying with TZD could likely have weakened the strength of evidence of the study. Notably, the median length of follow-up varied across different third-line medications. Therefore, it is possible that the risk of events could have been deduced from the data across a limited period of time. Our study results provide an important indication of the relative risks among third-line diabetes medications with respect to all-cause mortality, SH, and CVD. However, since the overall duration of this study was relatively short, additional studies are needed in estimating the long-term effect on cardiovascular events associated with DPP4i as one of the third-line glucose-lowering medications.

This retrospective population-based cohort study evaluated the risks of all-cause mortality, SH, and CVD events for patients intensifying with insulin, DPP4i, or TZD as a third-line medication. Among patients with T2DM on Met-SU dual therapy, DPP4i as a third-line medication had the lowest risk of all-cause mortality and SH events and posed no increased risk for CVD when compared to TZD and insulin. Intensification with insulin as a third-line medication had the greatest risk of mortality and SH events among the 3 glucose-lowering medications. There is a need for careful monitoring for baseline risks in patients intensifying with TZD and insulin, as well as careful assessment regarding the choice of a new third-line medication for T2DM management.

Supporting information

S1 STROBE Checklist. (DOCX)

S1 Table. List of relevant diagnosis codes. (DOCX)

S2 Table. Baseline characteristics of patient with DPP4i, insulin, or TZD as third-line medication after propensity-score weighting. (DOCX)

S3 Table. HR of all-cause mortality, SH, and CVD events for comparisons among DPP4i, insulin, TZD, and other glucose-lowering medications. (DOCX)

S4 Table. E-values for outcomes. (DOCX)

S5 Table. Adjusted HRs for the associations between the outcomes and observed baseline covariates.

(DOCX)

S6 Table. HR of all-cause mortality, SH, and CVD events using complete case analysis. (DOCX)

Acknowledgments

The authors wish to thank Mr Eric Tang and Garvin Cheng for statistical support, and acknowledge the Central Panel on Administrative Assessment of External Data Requests, Hong Kong Hospital Authority Head Office, for the provision of Hospital Authority data.

Author Contributions

Conceptualization: Carlos K. H. Wong.

Data curation: Carlos K. H. Wong, Emily T. Y. Tse.

Formal analysis: Carlos K. H. Wong, Kenneth K. C. Man, Chu Wa Ho.

Funding acquisition: Carlos K. H. Wong, Cindy L. K. Lam.

Investigation: Carlos K. H. Wong, Margaret Shi.

Methodology: Carlos K. H. Wong, Kenneth K. C. Man, Esther W. Chan, Chu Wa Ho, Ian C. K. Wong, Cindy L. K. Lam.

Project administration: Carlos K. H. Wong.

Resources: Carlos K. H. Wong.

Software: Carlos K. H. Wong, Kenneth K. C. Man, Chu Wa Ho.

Supervision: Carlos K. H. Wong, Cindy L. K. Lam.

Validation: Carlos K. H. Wong, Kenneth K. C. Man.

Visualization: Carlos K. H. Wong.

Writing - original draft: Carlos K. H. Wong, Margaret Shi, Chu Wa Ho.

Writing – review & editing: Carlos K. H. Wong, Esther W. Chan, Emily T. Y. Tse, Ian C. K. Wong, Cindy L. K. Lam.

References

- Gaster B, Hirsch IB. The Effects of Improved Glycemic Control on Complications in Type 2 Diabetes. JAMA internal medicine. 1998; 158(2):134–40. https://doi.org/10.1001/archinte.158.2.134 PMID: 9448551
- 7. Approaches to Glycemic Treatment. Diabetes Care. 2015; 38(Supplement 1):S41. https://doi.org/10. 2337/dc15-S010 PMID: 25537707
- Inzucchi SE, Bergenstal RM, Buse JB, Diamant M, Ferrannini E, Nauck M, et al. Management of hyperglycemia in type 2 diabetes, 2015: a patient-centered approach: update to a position statement of the American Diabetes Association and the European Association for the Study of Diabetes. Diabetes Care. 2015; 38(1):140–9. Epub 2014/12/30. https://doi.org/10.2337/dc14-2441 PMID: 25538310.
- National Institute for Health Health and Care Excellence (NICE). Type 2 Diabetes in Adults: Management: NICE guideline [NG28]. London: National Institute for Health and Care Excellence (UK), 2015
- Davies MJ, D'Alessio DA, Fradkin J, Kernan WN, Mathieu C, Mingrone G, et al. Management of Hyperglycemia in Type 2 Diabetes, 2018. A Consensus Report by the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD). Diabetes care. 2018; 41(12):2669– 701. Epub 2018/10/07. <u>https://doi.org/10.2337/dci18-0033</u> PMID: <u>30291106</u>; PubMed Central PMCID: PMC6245208.
- 9. Pharmacologic Approaches to Glycemic Treatment: Standards of Medical Care in Diabetes-2019. Diabetes Care. 2019; 42(Suppl 1):S90–s102. Epub 2018/12/19. <u>https://doi.org/10.2337/dc19-S009</u> PMID: 30559235.
- 7. Canadian Agency for Drugs and Technologies in Health. Optimal use recommendations for second and third-line therapy for patients with type 2 diabetes. CADTH optimal use report. 2013; 3(1d).
- Vijan S, Sussman JB, Yudkin JS, Hayward RA. Effect of patients' risks and preferences on health gains with plasma glucose level lowering in type 2 diabetes mellitus. JAMA Internal Medicine. 2014; 174 (8):1227–34. https://doi.org/10.1001/jamainternmed.2014.2894 PMID: 24979148
- Scheen JA. Cardiovascular Effects of New Oral Glucose-Lowering Agents: DPP-4 and SGLT-2 Inhibitors. Circulation Research. 2018; 122(10):1439–59. <u>https://doi.org/10.1161/CIRCRESAHA.117</u>. 311588 PMID: 29748368
- 10. Jil M, Rajnikant M, Richard D, Iskandar I. The effects of dual-therapy intensification with insulin or dipeptidylpeptidase-4 inhibitor on cardiovascular events and all-cause mortality in patients with type 2

diabetes: A retrospective cohort study. Diabetes Vasc Dis Re. 2017; 14(4):295–303. https://doi.org/10. 1177/1479164116687102 WOS:000403606900004. PMID: 28330386

- Yu CG, Fu Y, Fang Y, Zhang N, Sun RX, Zhao D, et al. Fighting type-2 diabetes: present and future perspectives. Current Medicinal Chemistry. 2017; 09:09. https://doi.org/10.2174/ 0929867324666171009115356 PMID: 28990512.
- Garber AJ, Abrahamson MJ, Barzilay JI, Blonde L, Bloomgarden ZT, Bush MA, et al. Consensus statement by the American Association of Clinical Endocrinologists and American College of Endocrinology on the comprehensive type 2 diabetes management algorithm–2018 executive summary. Endocrine Practice. 2018; 24(1):91–120. https://doi.org/10.4158/CS-2017-0153 PMID: 29368965
- Anyanwagu U, Mamza J, Mehta R, Donnelly R, Idris I. Cardiovascular events and all-cause mortality with insulin versus glucagon-like peptide-1 analogue in type 2 diabetes. Heart. 2016; 102(19):1581–7. Epub 2016/05/25. https://doi.org/10.1136/heartjnl-2015-309164 PMID: 27217068.
- Zaccardi F, Dhalwani NN, Dales J, Mani H, Khunti K, Davies MJ, et al. Comparison of glucose-lowering agents after dual therapy failure in type 2 diabetes: A systematic review and network meta-analysis of randomized controlled trials. Diabetes Obes Metab. 2018; 20(4):985–97. https://doi.org/10.1111/dom. 13185 WOS:000427114800023. PMID: 29205774
- Lozano-Ortega G, Goring S, Bennett HA, Bergenheim K, Sternhufvud C, Mukherjee J. Network metaanalysis of treatments for type 2 diabetes mellitus following failure with metformin plus sulfonylurea. Current medical research and opinion. 2016; 32(5):807–16. Epub 2015/12/25. https://doi.org/10.1185/ 03007995.2015.1135110 PMID: 26700585.
- Lee CM, Woodward M, Colagiuri S. Triple therapy combinations for the treatment of type 2 diabetes—A network meta-analysis. Diabetes research and clinical practice. 2016; 116:149–58. Epub 2016/06/21. https://doi.org/10.1016/j.diabres.2016.04.037 PMID: 27321330.
- Downes MJ, Bettington EK, Gunton JE, Turkstra E. Triple therapy in type 2 diabetes; a systematic review and network meta-analysis. PeerJ. 2015; 3:e1461. Epub 2015/12/15. <u>https://doi.org/10.7717/</u> peerj.1461 PMID: 26664803; PubMed Central PMCID: PMC4675096.
- Ou HT, Chang KC, Li CY, Wu JS. Comparative cardiovascular risks of dipeptidyl peptidase 4 inhibitors with other second- and third-line antidiabetic drugs in patients with type 2 diabetes. British journal of clinical pharmacology. 2017; 83(7):1556–70. Epub 2017/01/22. https://doi.org/10.1111/bcp.13241 PMID: 28109184; PubMed Central PMCID: PMC5465327.
- Man KKC, Coghill D, Chan EW, Lau WCY, Hollis C, Liddle E, et al. Association of Risk of Suicide Attempts With Methylphenidate Treatment. JAMA psychiatry. 2017; 74(10):1048–55. Epub 2017/07/27. https://doi.org/10.1001/jamapsychiatry.2017.2183 PMID: 28746699; PubMed Central PMCID: PMC5710471.
- Man KKC, Chan EW, Ip P, Coghill D, Simonoff E, Chan PKL, et al. Prenatal antidepressant use and risk of attention-deficit/hyperactivity disorder in offspring: population based cohort study. BMJ (Clinical research ed). 2017; 357:j2350. Epub 2017/06/02. https://doi.org/10.1136/bmj.j2350 PMID: 28566274; PubMed Central PMCID: PMC5450015.
- Wong AY, Wong IC, Chui CS, Lee EH, Chang WC, Chen EY, et al. Association Between Acute Neuropsychiatric Events and Helicobacter pylori Therapy Containing Clarithromycin. JAMA internal medicine. 2016; 176(6):828–34. Epub 2016/05/03. https://doi.org/10.1001/jamainternmed.2016.1586 PMID: 27136661.
- Law SWY, Lau WCY, Wong ICK, Lip GYH, Mok MT, Siu C-W, et al. Sex-Based Differences in Outcomes of Oral Anticoagulation in Patients With Atrial Fibrillation. Journal of the American College of Cardiology. 2018; 72(3):271–82. https://doi.org/10.1016/j.jacc.2018.04.066 PMID: 30012320
- White IR, Royston P, Wood AM. Multiple imputation using chained equations: Issues and guidance for practice. Statistics in Medicine. 2011; 30(4):377–99. https://doi.org/10.1002/sim.4067 PMID: 21225900
- 24. Suissa S. Lower Risk of Death With SGLT2 Inhibitors in Observational Studies: Real or Bias? Diabetes care. 2018; 41(1):6–10. https://doi.org/10.2337/dc17-1223 J Diabetes Care. PMID: 29263192
- Brookhart MA, Wyss R, Layton JB, Stürmer T. Propensity Score Methods for Confounding Control in Nonexperimental Research. Circulation: Cardiovascular Quality and Outcomes. 2013; 6(5):604–11. https://doi.org/10.1161/CIRCOUTCOMES.113.000359 PMID: 24021692
- Leyrat C, Seaman SR, White IR, Douglas I, Smeeth L, Kim J, et al. Propensity score analysis with partially observed covariates: How should multiple imputation be used? Statistical methods in medical research. 2019; 28(1):3–19. Epub 2017/06/03. https://doi.org/10.1177/0962280217713032 PMID: 28573919; PubMed Central PMCID: PMC6313366.
- Austin PC. An Introduction to Propensity Score Methods for Reducing the Effects of Confounding in Observational Studies. Multivariate Behavioral Research. 2011; 46(3):399–424. <u>https://doi.org/10.1080/00273171.2011.568786 PMID: 21818162</u>

- Haneuse S, VanderWeele TJ, Arterburn D. Using the E-Value to Assess the Potential Effect of Unmeasured Confounding in Observational Studies. Jama. 2019; 321(6):602–3. Epub 2019/01/25. https://doi.org/10.1001/jama.2018.21554 PMID: 30676631.
- VanderWeele TJ, Ding P. Sensitivity Analysis in Observational Research: Introducing the E-Value. Annals of internal medicine. 2017; 167(4):268–74. Epub 2017/07/12. <u>https://doi.org/10.7326/M16-2607</u> PMID: 28693043.
- Mathur MB, Ding P, Riddell CA, VanderWeele TJ. Web Site and R Package for Computing E-values. Epidemiology (Cambridge, Mass). 2018; 29(5):e45–e7. Epub 2018/06/19. https://doi.org/10.1097/ede. 00000000000864 PubMed Central PMCID: PMC6066405. PMID: 29912013
- Scheen AJ. Pharmacotherapy of 'treatment resistant' type 2 diabetes. Expert Opinion on Pharmacotherapy. 2017; 18(5):503–15. https://doi.org/10.1080/14656566.2017.1297424 PMID: 28276972
- Currie CJ, Poole CD, Evans M, Peters JR, Morgan CL. Mortality and other important diabetes-related outcomes with insulin vs other antihyperglycemic therapies in type 2 diabetes. The Journal of clinical endocrinology and metabolism. 2013; 98(2):668–77. Epub 2013/01/31. <u>https://doi.org/10.1210/jc.2012-3042 PMID: 23372169</u>.